This is an open access article under

© 2021. The Authors.

the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Turning the Orogenic Switch: Slab-Reversal in the Eastern Alps Recorded by Low-Temperature Thermochronology

Paul R. Eizenhöfer¹, Christoph Glotzbach¹, Lukas Büttner¹, Jonas Kley², and Todd A. Ehlers¹

¹Department of Geosciences, University of Tübingen, Tübingen, Germany, ²Geoscience Centre, University of Göttingen, Göttingen, Germany

Abstract Many convergent orogens, such as the eastern European Alps, display an asymmetric doubly vergent wedge geometry. In doubly vergent orogens, deepest exhumation occurs above the retrowedge. Deep-seismic interpretations depict the European plate dipping beneath the Adriatic, suggesting the pro-wedge location on the north side of the orogen. Our new thermochronometer data across the Eastern Alps confirm distinct shifts in the locus of exhumation associated with orogen-scale structural reorganizations. Most importantly, we find a general Mid-Miocene shift in exhumation (in the Tauern Window and the Southern Alps) and focus of modern seismicity across the Southern Alps. Taken together, these observations suggest a subduction polarity reversal at least since the Mid-Miocene such that the present-day pro-wedge is located on the south side of the Alps. We propose a transient tectonic state of a slow-and-ongoing slab reversal coeval with motion along the Tauern Ramp, consistent with a present-day northward migration of drainage divides.

Plain Language Summary When tectonic plates collide, they bend downwards and form two lithospheric wedges dipping in opposite directions, such as in the Eastern Alps. We present new crustal cooling data along a transect in the Eastern Alps confirming that surface rocks across the central Tauern Window originated from the deepest structural levels along the transect. South of the Tauern Window rocks were exhumed from higher depths compared to those north of it and were exhumed more recently, while seismic activity is also focused across the Southern Alps. These observations suggest a subduction polarity reversal because they are inconsistent with the original southern and northern locations of overriding and subducting plates, respectively, >15 million years ago. This interpretation is contrary to lithosphere-scale tomography that shows no change in subduction polarity. Therefore, we propose a transient tectonic state, that is, a slow-and-ongoing subduction polarity reversal that initiated when Tauern Window rocks began their steep ascent to the surface along a deep-seated fault known as the Tauern Ramp. This study bridges observations in the mantle, crust and on the surface over geologic time.

1. Introduction

Subduction polarity reversals have been suggested in several orogens, including Taiwan (Teng et al., 2000), the Solomon Islands arc (Cooper & Taylor, 1985; Petterson et al., 1999), the European Alps (Lippitsch et al., 2002, 2003), and also in the Caledonide (Dewey, 2005) and Pamir orogens (Kufner et al., 2016). Seismic tomography results from the eastern European Alps have led to the hypothesis that a change in subduction polarity also occurred there (Handy et al., 2015; Kissling et al., 2006; Lippitsch, 2002; Lippitsch et al., 2003; Luth et al., 2013). The focus of this study is the effect of a "switch" in the direction (or polarity) of subduction on upper plate deformation, exhumation, and topography. Many convergent orogens can be described as "doubly vergent", such as in the European Alps (Argand, 1916), whereby pro- (material above the subducting plate) and retro- (material overlying the overriding plate) orogenic wedges develop (Willett et al., 1993). The pro- and retro-wedges develop as two critically tapered Coulomb wedges in response to a balancing of forces and provide a useful framework for understanding the evolution of deformation, exhumation, and topography in these settings (e.g., Davis et al., 1983; Beaumont et al., 1996; Willett & Brandon, 2002).

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL092121

Key Points:

- Thermochronologic data in the Eastern Alps is consistent with a transient tectonic state toward complete slab reversal
- The pro-wedge has switched from north to south of the Periadriatic Fault along TRANSALP
- Mid-Miocene motion along the Tauern Ramp is the consequence of slab-reversal

Supporting Information:

- Supporting Information S1
- Table S1
- Table S2 Table S3
- Table S4

Correspondence to:

P. R. Eizenhöfer, paul-reinhold.eizenhoefer@unituebingen.de

Citation:

Eizenhöfer, P. R., Glotzbach, C., Büttner, L., Kley, J., & Ehlers, T. A. (2021). Turning the orogenic switch: Slab-reversal in the Eastern Alps recorded by low-temperature thermochronology. Geophysical Research Letters, 48, e2020GL092121. https://doi.org/10.1029/2020GL092121

Received 17 DEC 2020 Accepted 31 JAN 2021

ADVANCING **EARTHAND** SPACE SCIENCE





Geophysical Research Letters



Figure 1. Crustal responses to proposed lower lithospheric slab-reversal in the eastern European Alps. (a) Geological map with new AHe and ZHe ages (in Ma) (b) Seismicity recorded by the International Seismological Center (ICS; Storchak et al., 2017) and the GEOFON Data Center (1993) over the past 10 years (GFZ), and drainage divide migration trends based on χ -analyses (Robl et al., 2017; Winterberg & Willett, 2019; divides outlined as blue lines; see also Text S3). (c) to (e) Best-fit t-T-path models of samples situated north of the INN ("Northern Calcareous Alps"), between the INN and the PF ("Tauern Window") and south of the PF ("Southern Alps"); see Text S2 and Table S4 for details. PF, Periadriatic Fault; VAL, Valsugana thrust system; MTL, Montello thrust; INN, Inntal fault; SEMP, Salzach-Ennstal-Mariazell-Puchberg fault; GD, Giudicarie fault; \dagger , mixed age (detrital or partially reset); *, single grain age.

If a subduction polarity reversal takes place, the double-wedge system should reflect this change which could be observed through: (i) changes in the location of deepest exhumation in the retro-wedge, (ii) sequential fault activity in the "new" pro-wedge, and (iii) the location of the main drainage divide that separates minimum from maximum tapered topography along the orogenic wedges (e.g., Beaumont & Quinlan, 1994; Beaumont et al., 1996, Willett et al., 1993, 2001; Figure 1). If the Eastern Alps experienced a subduction polarity reversal, then flipping of the pro- and retro-wedges should be recorded in upper lithospheric deformation, exhumation, and topography. Recent seismic imaging of the Alpine Moho (Kästle et al., 2020; Kissling et al., 2006; Lippitsch, 2002; Lippitsch et al., 2003, and references therein) document the present-day structure of the lower lithosphere (Figure 2) and highlight the presence of two slab remnants beneath the Central and Eastern Alps. These slabs are located approximately along the Periadriatic Fault (PF; Figure 1a) at ~135–165 km depth. However, the slab geometry beneath the TRANSALP geophysical transect in the Eastern Alps (Lüschen et al., 2004, 2006) in the vicinity to the "Moho gap" (Figures 1a and 1b; Hetényi et al., 2018; Spada et al., 2013) has led to different interpretations invoking either a southward (Castellarin, Nicolich,

Geophysical Research Letters





Figure 2. Cross-sectional N-S view of a 20 km swath along TRANSALP. (a) Distribution of thermochronologic ages. Green, dark blue, and light blue dashed lines outline proposed age trends (ZHe, AFT, and AHe, respectively), vertical gray lines correspond to major fault locations. (b) Topographic profile with locations of samples and published thermochronology data. (c) Locations of the European and Adriatic Moho based on receiver function analyses (Kummerow et al., 2004), simplified structural geometry and seismicity (GFZ, GEOFON Data Center, 1993; ICS, Storchak et al., 2017). Note the potentially continued southward shift of present-day fault activity south of the MTL (dashed, black line; Anselmi et al., 2011; Serpelloni et al., 2016). Arrows illustrate the general direction of displacement and relative magnitude of exhumation since continental collision. Fault acronyms as in Figure 1. AFT/ZFT data from Grundmann and Morteani (1985), Coyle (1994), Stöckert et al. (1999), Trautwein et al. (2001), Steenken et al. (2002); Most-Angelmeier (2003), Zattin et al. (2003, 2006) and Betrand et al. (2017).

et al., 2006; Lammerer et al., 2008) or northward (Handy et al., 2015; Kissling et al., 2006; Schmid et al., 2004) polarity of continental subduction. The latter case would require that the Adriatic plate is situated beneath the European, implying a slab reversal at some time since the onset of Eocene continental collision.

Here, we evaluate the hypothesis that a switch in subduction polarity occurred under the Eastern Alps. We do this using new and existing records of rock cooling and exhumation histories from low-temperature thermochronology along TRANSALP, and existing geophysical and topographic observations (Figures 1 and 2). Our results indicate a slow-and-ongoing reversal since at least the Mid-Miocene and may facilitate the identification of slab reversals in similar tectonic settings.

2. Geological Background

Continental collision in the Eastern Alps resulted in multiple tectonic events since the Eocene. In recent reconstructions (e.g., Handy et al., 2015; Schmid et al., 2004, 1996), initial southward oceanic subduction of the European slab focused upper lithospheric deformation north of the PF and ceased in the Eocene after contact between the European and Adriatic plates. Continental collision in the Oligocene led to upper lithospheric re-organization and the formation of retro-thrusts in the Southern Alps ("Pre-Adamello Phase"; Doglioni, 1992; Castellarin, Vai, & Cantelli, 2006). This produced a doubly vergent orogen with "pro" and "retro" wedges on the north and south sides of the orogen, respectively. During this time the Tauern Window retrograded from peak-metamorphic conditions ("Tauernkristallisation"; e.g., Favaro et al., 2015). Intrusions along the PF are interpreted to be the result of an Early Oligocene slab break-off along the European Plate (Blanckenburg & Davies, 1995). This is followed by a general relocation of deformation from north to south of the PF since the Late Oligocene (e.g., Handy et al., 2015; Schmid et al., 2004, 1996) despite some post-Eocene shortening across the Northern Calcareous Alps (Auer & Eisbacher, 2003). By the Early Miocene, the front of the advancing Alpine nappe stack had reached the Subalpine Molasse and came to a halt in Switzerland (Burkhard & Sommaruga, 1998), while reverse faulting occurred south of the PF (Handy et al., 2015). The indentation of Adria into Europe resulted in lateral extrusion tectonics in the Eastern Alps during the Miocene (Frisch et al, 1998, 2000; Ratschbacher et al., 1991; Rosenberg et al., 2018; Scharf et al., 2013), coinciding with strike-slip motion and thrust activity along E-W and N-S striking sections of the PF, respectively (Bartosch et al., 2017; Mancktelow et al., 2001; Müller et al., 2001; Pleuger et al., 2012; Wölfler et al., 2011; Zwingmann & Mancktelow, 2004, and references therein). Since the Mid-Miocene, active shortening along in-sequence thrust systems occurred south of the PF (Castellarin & Cantelli, 2000; Castellarin, Nicolich, et al., 2006; Schönborn, 1992, 1999) in contrast to a general absence of shortening north of the PF. Vertical orogenic growth took place throughout the Miocene as evidenced by Mid-Miocene rapid exhumation of core complexes along deep-seated, crustal-scale thrusts in the Central and Eastern Alps. Late Tauern Window exhumation is associated with displacement along the S-dipping Tauern Ramp (Bertrand et al, 2015, 2017; Rosenberg et al., 2018), whereas exhumation of the Lepontine Dome in the Central Alps is linked to displacement along the N-dipping PF (Herwegh et al., 2017; Vernon et al., 2009). However, across-strike orogenic growth is most evident in the Central to Western Alps as recorded by southward propagation of deformation beneath the Po Plain (Pieri & Groppi, 1981; Schönborn, 1992). The northern deformation front was revived in the Central Alps during Mid-Miocene to Pliocene thrusting of the Jura Mountains (Burkhard & Sommaruga, 1998; Schmid et al., 1996).

Present-day geophysical and geomorphic observations in the Eastern Alps provide additional insight into the present plate tectonic configuration of the Alpine system. Seismicity is most abundant across the Southern Alps at the Montello thrust, and less abundant north of the Tauern Window (Figures 1b and 2c). Shallow crustal (<20 km) seismicity beneath the Venetian Alps across the Montello thrust (e.g., Jozi Najafabadi et al., 2020) indicate buried south-verging thrusts (Anselmi et al., 2011; Serpelloni et al., 2016) and a high seismic potential (Barba et al., 2013; Galadini et al., 2005; Moratti et al., 2019). Deep-seismic reflection images and receiver function analyses in the Eastern Alps along the TRANSALP and EASI geophysical profiles (Gebrande et al., 2002; Hetényi et al., 2018; Kummerow et al., 2004; Lüschen et al., 2004) clearly denote south- and northward verging lower lithospheric European and Adriatic slabs (Figure 2c). Both characteristics confirm the doubly vergent architecture of the Alps (Argand, 1916). High-resolution seismic interpretations along EASI by Hetényi et al. (2018) imaged the Adriatic slab dipping beneath the European, consistent with previous works (Babuška et al., 1990; Karousova et al., 2013). Further to the SE the Dinaridic section of the Adriatic plate is subducting beneath the Dinarides since the late Cretaceous (e.g., Andrić et al., 2018; and references therein). Based on lower resolution tomography along TRANSALP, the precise relationship between the European and Adriatic plates, as well as the location of the Moho and geometry of overriding versus subducting slabs (Gebrande et al., 2002; Kästle et al., 2020; Kummerow et al., 2004; Lippitsch, 2002; Lippitsch et al., 2003; and references therein), remains elusive, especially in the vicinity of the "Moho gap" (Hetényi et al., 2018; Spada et al., 2013).

Geomorphic studies on the mobility of present-day drainage divides (Robl et al., 2017; Winterberg & Willett, 2019) indicate that the Alpine landscape remains in disequilibrium and highlight a general northward migration of divides, especially in the transition zone from the Central to Eastern Alps (Figure 1b). These



trends in divide migration were interpreted not as a consequence of glaciation but rather changes in base levels in basins surrounding the Alpine system (e.g., the Pannonian Basin and Mediterranean Sea). However, potential links to changes in lower lithospheric geometries have not been evaluated.

3. Methodology and Results

Low-temperature thermochronology provides insights into the cooling history of rocks in the upper lithosphere (e.g., Braun, 2003; McQuarrie & Ehlers, 2015, 2017; Willett et al., 2020). Erosion during mountain building is one mechanism for rock cooling as mass is advected (exhumed) toward the Earth's surface (e.g., Ehlers, 2005). Thermochronometer data record the time since a rock passes through an effective closure temperature (e.g., Reiners et al., 2005). Previous work suggests closure temperatures of 180–250°C for the zircon fission track (ZFT) system (e.g., Bernet, 2009), 140–220°C for the zircon (U-Th)/He (ZHe) system (Guenthner et al., 2013), 80–120°C for the apatite fission track (AFT) system (e.g., Ketcham et al., 2007) and 50–70°C for the apatite (U-Th/He) (AHe) system (e.g., Farley, 2000; Flowers et al., 2009) corresponding to upper lithospheric depths of ~12–2 km assuming common geothermal gradients. Residence above the closure temperatures (e.g., following continental subduction and/or subsidence) resets thermochronologic ages, whereas partially reset ages may occur in rocks that were heated to temperatures close to the specific closure temperature. Our new thermochronometer data (Tables S1–S4) are densely spaced along the length of TRANSALP (Figures 1 and 2; Text S1 and S2) and complemented by existing data.

Two general trends in cooling ages are observable along TRANSALP (Figure 2a). First, ZFT and ZHe data in the Tauern Window reflect the location of the deepest levels of exhumation across the transect. Reset ZFT ages across the Tauern Window range from 30 to 10 Ma (Bertrand et al., 2017) implying >10 km of exhumation since at least 30 Ma. This observation is consistent with previous studies (v. Blanckenburg et al., 1989; Fügenschuh et al., 1997; Lammerer et al., 2008; Selverstone et al., 1995) suggesting deformation over the same period until the Pliocene as constrained by our youngest, reset AHe ages at ~6 Ma (Figures 1a, 1c and 2a) and supported by our time-temperature (t-T)-path inversions. High cooling rates across the Tauern Window, sensu lato (i.e., the area between the Inntal fault and the PF), occur from the Middle to Late Miocene (Figure 1d; Text S2 and Table S4). In contrast, ZHe ages north and south of the Tauern Window are mostly unreset (>300 Ma) and become successively younger near the Tauern Window. Deep exhumation of >10 km is, thus, strictly limited to the Tauern Window as bedrock to the north and south did not cross isotherms that would completely reset the ZHe system during the Alpine orogeny. The symmetric "U-shape" distribution and systematic continuity in the distribution of thermochronometer ages when transitioning from the Tauern Window to the Southern Alps (Rosenberg et al., 2018), underlined by our new ZHe and AHe data (Figure 2a), does not require significant (i.e., >1-2 km) reverse faulting along the PF, and can readily be explained by displacement along the Tauern Ramp (Lüschen et al., 2004, 2006) during Miocene time (Lammerer et al., 2008; Ortner et al., 2006).

Second, AFT and AHe data show a similar "U-shape" distribution across the Tauern Window but have also been reset in the Northern Calcareous Alps and Southern Alps. Resetting occurred during the Alpine orogeny and the AFT ages reveal higher degrees of exhumation and younger active exhumation in the south (i.e., between 3 and 6 km exhumation) relative to the north since initial collision between the European and Adriatic plates. Across most of the Northern Calcareous Alps and Northern Alpine Molasse, the AHe system yields ages of 7-28 Ma, whereas the youngest ages (<10 Ma) north of the Tauern Window are restricted to the vicinity of the Inntal fault and potentially related to transpressional tectonics. An AFT age of 31 Ma may indicate overall shallower levels of exhumation (i.e., <3 km) since the Eocene and, importantly, during the time window of the proposed slab reversal. In contrast, recorded AFT ages in the Southern Alps (Zattin et al., 2003, 2006) range from 8 to 24 Ma (with AHe ages expected to be younger), and the AHe system yielded ages between ~20 and 12 Ma to the north and south of the Dolomites, respectively (Figure 2a). These observations are additionally confirmed by our t-T-path models, which suggest not only that the Southern Alps (Figure 1e) witnessed higher temperatures relative to the Northern Calcareous Alps (Figure 1c) since the Eocene. Our t-T-path models indicate multiple episodes of increased cooling from higher temperatures in the Southern Alps in contrast to relatively slow cooling at lower temperatures north of the Inntal Fault (Figures. 1c vs. 1e), that is, at ~23-20 Ma and at ~15-10 Ma. Proposed timing of fault activity along TRANSALP (e.g., Lüschen et al., 2004) suggests Mid-Miocene to Pliocene and Late Oligocene to Late







Figure 3. Conceptual model for the evolution of the lower lithospheric slab polarity reversal along TRANSALP in the Eastern Alps. (a) Initial doubly vergent orogen geometry prior to reversal (Willett & Brandon, 2002; Willett et al., 1993). (b) Initiation of slab reversal coeval with Tauern Ramp activity and overthrusting of the Adriatic plate. (c) Re-organization of the lower lithospheric subduction geometry, upper lithospheric fault activity and topographic response following the indentation of the Adriatic plate.

Miocene fault activity in the Southern Alps and the Northern Calcareous Alps, respectively. Hence, the southward migration of increased cooling in the Southern Alps is likely associated with in-sequence fault activity that took place prior to, and during rapid cooling across the Tauern Window (Figure 1d). Present-day active faulting in the southernmost section of TRANSALP across the Montello thrust (Anselmi et al., 2011; Serpelloni et al., 2016) appears to indicate a continuation of the southward migration of fault activity.

4. Discussion

Mechanical models for tectonics in doubly vergent orogens predict the deepest levels of exhumation occur in the retro-wedge side of an orogen as bedrock that is exposed there was sourced from the deepest crustal levels during continental collision (Figure 3a; Willett et al., 1993, 2001; Willett & Brandon, 2002). This relationship has been observed, for example, across the Southern Alps of New Zealand (Batt et al., 2000), the Olympic Mountains (Batt et al., 2001; Michel et al., 2018), Taiwan (Liu et al., 2001), as well as the Patagonian Andes (Fosdick et al., 2013). Strong climatic gradients may affect this exhumation pattern (Willett et al., 1993) but are negligible across-strike in the European Alps as moisture transport dominantly follows an Atlantic W-to-E trajectory since at least the Pliocene (e.g., Botsyun et al., 2020). Knowledge of the location of the deepest exhumation in a convergent orogen, hence potentially that of the retro-wedge, would be reflective of subduction polarity.

We interpret the presence of reset thermochronometric ages across the western Tauern Window to represent the present locus of deepest exhumation along TRANSALP and indicative of a retro-wedge position of the Eastern Alps north of the PF. Orogen-parallel extrusion tectonics amounting to ~50 km of E-W extension (e.g., Rosenberg et al., 2018; Wolff et al., 2020) are not expected to affect the fundamental N-S oriented doubly vergent wedge mechanical framework in the Eastern Alps. The Tauern Ramp therefore represents a deep-seated back-thrust system active from at least the Mid-Miocene to Pliocene based on our t-T-path models (Figure 1d) and the youngest AHe ages (Figures 1a and 2a). This scenario is consistent with the Early Miocene initiation of exhumation of the Tauern Window post peak-metamorphic conditions during the Oligocene "Tauernkristallisation" (Schmid et al., 2013; Favaro et al., 2015), a general southward stepping of fault activity and youngest episodes of increased cooling rates in the Southern Alps (Figure 1e), and low cooling rates within the Northern Calcareous Alps (Figure 1c). Rapid Miocene exhumation of the Tauern Window can be interpreted as retro-wedge

deformation consistent with fault-propagation folding on the largely blind, crustal-scale Tauern Ramp (Ortner et al., 2006) with a maximum shortening of ~14–17 km fed into the basal thrust of the northern wedge (Lammerer et al., 2008). Earlier periods of fast cooling in the Southern Alps, prior to the time of Tauern Ramp activity, may be either (i) associated with the original retro-wedge position of the Southern Alps and displacement along the Dolomite Ramp, or (ii) interpreted as an incipient pro-wedge response of the Southern Alps to the Oligocene phase of Tauern Window exhumation not linked to Tauern Ramp activity (Figure 3a).

The timing for a switch between pro- and retro-wedge positions, therefore, is structurally most reconcilable with the onset of motion along the Tauern Ramp. Earlier exhumation across the Tauern Window prior to activity along the Tauern Ramp likely took place in a pro-wedge position potentially associated with prior

duplex formation (e.g., Lammerer et al., 2008). Following this interpretation, shallow exhumation along back-thrust systems, sometimes reactivating pre-existing faults (e.g., Auer & Eisbacher, 2003), took place in the Northern Calcareous Alps during the Mid-to Late Miocene as topography developed above the Tauern Ramp (e.g., Frisch et al., 1998). Consequently, the Southern Alps developed into a pro-wedge not later than after the initiation of Tauern Window exhumation along the Tauern Ramp since the Mid-Miocene (Figure 3b). The Southern Alps respond to their new role as a pro-wedge through increased fault activity, southward thrust front propagation and lengthening of the wedge (Figure 3c; Castellarin & Cantelli, 2000; Castellarin, Nicolich, et al., 2006; Petricca et al., 2019; Zattin et al., 2003, 2006). The youngest phase of rapid cooling at ~15-10 Ma in the Southern Alps is observed at the southernmost section of TRANSALP (Figure 1e). This observation is corroborated by the in-sequence activation of the Valsugana and Montello thrust systems since the Mid-Miocene (Castellarin & Cantelli, 2000; Castellarin, Nicolich, et al., 2006).

The relocation of deformation within the orogenic system serves to (i) accommodate within the subducting plate an increase in strain above a detached lower lithospheric slab, (ii) decrease its taper to a new critical state by adding imbricate thrust sheets at the deformation front as active deformation shifts sequentially southwards, and (iii) balance the accretionary influx to re-establish exhumational steady-state. These lines of thought are in agreement with present-day seismic and kinematic observations. For example, present-day seismic activity is predominantly centered to the south of the Alps (Figures 1b and 2c; Danesi et al., 2015; Petricca et al., 2019). GPS-based kinematics of upper lithospheric deformation (Nocquet & Calais, 2004) suggest the regions of highest strain are at the southernmost extent of TRANSALP. A high seismic potential has been associated with buried south-verging thrust sheets across the Montello thrust (Moratto et al., 2019; Serpelloni et al., 2016) indicating a continued, sequential migration of south-verging thrust faulting.

Several lines of evidence suggest that subduction polarity has switched along TRANSALP (Figures 1 and 2) since the Mid-Miocene. Contrary to the thermochronologic, structural and present-day seismicity record, deep geophysical images show no clear evidence of a subduction polarity reversal (Figure 2c; Kästle et al., 2020; Kummerow et al., 2004, and reference therein). The region between the Adriatic and European plates beneath TRANSALP has a low velocity contrast ("slab gap," Handy et al., 2015) and no definite subduction polarity (Lippitsch, 2002; Lippitsch et al., 2003). The ongoing indentation of the Adriatic plate (Figure 3c), however, provides a tectonic mechanism that is able to initiate a subduction polarity reversal while remaining consistent with: (i) recent deep seismic images if interpreted as a transient lithospheric state toward complete slab reversal (Figure 2c), and (ii) the thermochronologic, structural and present-day earthquake record (Figures 1 and 2) that are best explained by a switch between pro- and retro-wedges. Continued southward subduction without invoking a slab reversal (Castellarin, Nicolich, et al., 2006; Lammerer et al., 2008), could only be reconciled with a switch between pro- and retro-wedges in a decoupled lithosphere architecture where the Adriatic plate is wedging into European crust as proposed for the Central Alps (e.g., Rosenberg & Kissling, 2013). However, the lack of evidence for Moho overlaps along both, TRANSALP (Kummerow et al., 2004) and EASI (Hetényi et al., 2018), favors a polarity change of lithospheric mantle subduction. Seismic tomography results beneath EASI, ~100 km east of TRANSALP, are interpreted to show a steeply dipping Adriatic slab. Higher resolution seismic tomography is required to resolve the relative coupling of crust and lithospheric mantle.

Interpreting the main drainage divide as the point between minimum and maximum critically tapered topography can be used to define the surface expression of pro- and retro-wedges (Figure 3a; Willett et al., 1993, 2001). Slab reversal would induce a transient period of landscape disequilibrium expressed through divide migration toward the new retro-wedge (Figure 3b). Northwards migrating divides in the Central to Eastern Alps (Figure 1b; Robl et al., 2017; Winterberg & Willett, 2019) are supportive of this line of thought and may, hence, represent the transient response to an ongoing switch from initially southward to presently northward directed continental subduction along TRANSALP.

Multiple factors control the occurrence and timing of slab reversal in continent-continent collisional settings such as material properties, maturity of the orogen (i.e., internal temperature regimes) and external stress fields (Beaumont et al., 1996; Dewey, 2005; Houseman et al., 1981, 2000; Luth et al., 2013; Pysklywec et al., 2000; Pysklywec, 2001). Numerical models that explore the conditions for slab reversal (Pysklywec, 2001) indicate the onset of slab reversal after indentation of the retro-into the pro-lithospheric mantle leads to slab break-off of the subducting slab given sufficiently low geothermal gradients. The Eastern Alps

ons) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons Licens

19448007, 2021, 6, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2020GL092121 by University Of Glasgow, Wiley Online Library on [11/01/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.

may have provided favorable conditions that promoted lower lithospheric slab reversal in a continent-continent collisional setting during the Miocene. Oligocene slab break-off may have raised the European lower lithosphere toward shallower levels causing not only regional surface uplift above the subducting plate but also enabling Adria to indent into and initiate subduction beneath Europe. The Adriatic plate (or indenter) is generally regarded as more rigid than the thickened European plate as quartz-rich basement rocks in the latter face lower lithosphere in the former. At upper lithospheric levels, the Austroalpine units experienced lower degree metamorphism in contrast to the higher degree metamorphic Penninic units across the Tauern Window (e.g., Handy et al., 2015; Scharf et al., 2013). This rheologic contrast between Europe and Adria perhaps continues at depth despite the Oligocene intrusions along the PF (e.g., Willingshofer & Cloetingh, 2003; Rosenberg, 2004). Slowing plate tectonic convergence between Europe and Adria from initially ~7.5 to ~5 mm/yr since ~20 Ma (Schmid et al., 1996; Müller et al., 2019) may have contributed to prolonging the process of slab reversal. Thus, we suggest that the Mid-Miocene initiation of slab reversal is still in progress while Moho signatures along TRANSALP are interpreted to reflect the ongoing indentation of Adria into Europe (Figures 2c and 3b; Gebrande et al., 2002; Kummerow et al., 2004). This interpretation is in line with recent interpretations of seismic tomography (Kästle et al., 2020) where Adriatic subduction is suggested to be minor.

5. Conclusions

Recent interpretations of seismic data along TRANSALP in the Eastern Alps (Figure 2c; Gebrande et al., 2002; Kummerow et al., 2004; Lüschen et al., 2004, 2006) confirmed a doubly vergent orogen geometry (Argand, 1916; Willett et al., 1993, 2001; Willett & Brandon, 2002). The across-strike distribution of thermochronometer ages identify the location of deepest levels of exhumation across the Tauern Window and favor a present-day southern and northern location of the pro- and retro-wedges, respectively. This interpretation entails a reversal in subduction polarity at least since the onset of motion along the Tauern Ramp in the Mid-Miocene. Northward migration of drainage divides, kinematic and seismic observations (Figures 1 and 2) not only support this line of arguments but also suggest a slow-and-ongoing reversal accompanying the indentation of European mantle lithosphere by the Adriatic plate (Figure 3c). The proposed scenario is in contrast to a fast-and-completed Miocene slab reversal (e.g., Handy et al., 2015) and explains the lack of a clear subduction polarity in geophysical images (Kummerow et al., 2004). This tectonic state along TRANSALP may represent a transition between the Central Alps and the Eastern Alps. In the former, where the Lepontine Dome would demark the location of the retro-wedge position prior to the onset of European slab rollback (Schlunegger & Kissling, 2015), slab reversal did not occur, whereas the process of subduction polarity reversal is potentially completed in the latter (Kästle et al., 2020). Higher resolution tomographic images beneath TRANSALP, shedding more light onto the lower lithospheric slab geometries, that is, ongoing indentation versus full slab reversal, would strengthen the here proposed mantle-to-surface link between slab reversal and its long-to short-term hinterland denudational response in continent-continent collisional settings.

Data Availability Statement

Our new thermochronology data can be accessed from the GFZ Data Services (https://doi.org/10.5880/fidgeo.2020.048). Seismic data were taken from the International Seismological Center (Storchak et al., 2017) and the GEOFON Data Center (1993). Digital elevation models were produced by the U.S./Japan ASTER Science Team (2000). Geomorphic analyses were performed through "TopoToolbox V2" (Schwanghart & Scherler, 2014). Receiver function analysis was adopted from Kummerow et al. (2004).

References

Andrić, N., Vogt, K., Matenco, L., Cvetković, V., Cloetingh, S. A. P. L., & Gerya, T. (2018). Variability of orogenic magmatism during Mediterranean-style continental collisions: A numerical modeling approach. Gondwana Research, 56, 119-134. https://doi.org/10.1016/j.gr.2017.12.007

Anselmi, M., Govoni, A., De Gori, P., & Chiarabba, C. (2011). Seismicity and velocity structures along the south-Alpine thrust front of the Venetian Alps (NE-Italy). Tectonophysics, 513(1-4), 37-48. https://doi.org/10.1016/j.tecto.2011.09.023

Argand, E. (1916). Sur l'arc des Alpes occidentales: Ecologae Geologicae Helvetiae, v. 14, pp. 145–191.

19448007, 2021, 6, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2020GL092121 by University Of Glasgow, Wiley Online Library on [11/01/2023], See the Terms and Conditions (https://onlinelibrary.wiley

Acknowledgments

This study was supported by the German Research Foundation (DFG, GL724/8-1 to C.G., KL 495/28-1 to J.K., and EH329/19-1 to T.A.E.) under the priority program 4D-MB and is a contribution to the AlpArray initiative. This manuscript benefitted from discussions with Mark Handy. The authors thank the Editor Lucy Flesch and three anonymous Reviewers for constructive comments that greatly improved the manuscript.

nd-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons Licens



Auer, M., & Eisbacher, G. H. (2003). Deep structure and kinematics of the Northern Calcareous Alps (TRANSALP profile). International Journal of Earth Sciences, 92, 210–227. https://doi.org/10.1007/s00531-003-0316-0

- Babuška, V., PlomerovÁ, J., & Granet, M. (1990). The deep lithosphere in the Alps: A model inferred from P residuals. *Tectonophysics*, *176*, 137–165. https://doi.org/10.1016/0040-1951(90)90263-8
- Barba, S., Finocchio, D., Sikdar, E., & Burrato, P. (2013). Modeling the interseismic deformation of a thrust system: Seismogenic potential of the Southern Alps. *Terra Nova*, 25(3), 221–227. https://doi.org/10.1111/ter.12026
- Bartosch, T., Stüwe, K., & Robl, J. (2017). Topographic evolution of the Eastern Alps: The influence of strike-slip faulting activity. Lithosphere, 9, 384–398. https://doi.org/10.1130/L594.1
- Batt, G. E., Brandon, M. T., Farley, K. A., & Roden-Tice, M. (2001). Tectonic synthesis of the Olympic Mountains segment of the Cascadia wedge, using two-dimensional thermal and kinematic modeling of thermochronological ages. *Journal of Geophysical Research*, 106, 26731–26746. https://doi.org/10.1029/2001JB000288
- Batt, G. E., Braun, J., Kohn, B. P., & McDougall, I. (2000). Thermochronological analysis of the dynamics of the Southern Alps, New Zealand. *The Geological Society of America Bulletin*, 112, 250–266. https://doi.org/10.1130/0016-7606(2000)112<250:TAOTDO>2.0.CO;2 Beaumont, C., Ellis, S., Hamilton, J., & Fullsack, P. (1996). Mechanical model for subduction-collision tectonics of Alpine-type compres
 - sional orogens. Geology, 24, 675–678. https://doi.org/10.1130/0091-7613(1996)024<0675:MMFSCT>2.3.CO;2
- Beaumont, C., & Quinlan, G. (1994). A geodynamic framework for interpreting crustal-scale seismic-reflectivity patterns in compressional orogens. *Geophysical Journal International*, 116, 754–783. https://doi.org/10.1111/j.1365-246X.1994.tb03295.x
- Bernet, M. (2009). A field-based estimate of the zircon fission-track closure temperature. *Chemical Geology*, 259, 181–189. https://doi. org/10.1016/j.chemgeo.2008.10.043
- Bertrand, A., Rosenberg, C., & Garcia, S. (2015). Fault slip analysis and late exhumation of the Tauern Window, Eastern Alps. *Tectonophysics*, 649, 1–17. https://doi.org/10.1016/j.tecto.2015.01.002

Bertrand, A., Rosenberg, C., Rabaute, A., Herman, F., & Fügenschuh, B. (2017). Exhumation mechanisms of the Tauern Window (Eastern Alps) inferred from apatite and zircon fission track thermochronology. *Tectonics*, 36, 207–228. https://doi.org/10.1002/2016TC004133 Botsyun, S., Ehlers, T. A., Mutz, S. G., Methner, K., Krsnik, E., & Mulch, A. (2020). Opportunities and challenges for paleoaltimetry in "small"

orogens: Insights from the European Alps. Geophysical Research Letters, 47, e2019GL086046. https://doi.org/10.1029/2019GL086046

- Braun, J. (2003). Pecube: A new finite-element code to solve the 3D heat transport equation including the effects of a time-varying, finite amplitude surface topography. *Computational Geosciences*, 29, 787–794. https://doi.org/10.1016/S0098-3004(03)00052-9
- Burkhard, M., & Sommaruga, A. (1998). Evolution of the western Swiss Molasse basin: Structural relations with the Alps and the Jura belt. *Geological Society, London, Special Publications, 134*(1), 279–298. https://doi.org/10.1144/GSL.SP.1998.134.01.13
- Castellarin, A., & Cantelli, L. (2000). Neo-Alpine evolution of the Southern Eastern Alps. Journal of Geodynamics, 30, 251–274. https://doi.org/10.1016/S0264-3707(99)00036-8
- Castellarin, A., Nicolich, R., Fantoni, R., Cantelli, L., Sella, M., & Selli, L. (2006). Structure of the lithosphere beneath the Eastern Alps (southern sector of the TRANSALP transect). *Tectonophysics*, 414, 259–282. https://doi.org/10.1016/j.tecto.2005.10.013
- Castellarin, A., Vai, G. B., & Cantelli, L. (2006). The Alpine evolution of the Southern Alps around the Giudicarie faults: A Late Cretaceous to Early Eocene transfer zone. *Tectonophysics*, 414(1–4), 203–223. https://doi.org/10.1016/j.tecto.2005.10.019
- Cooper, P. A., & Taylor, B. (1985). Polarity reversal in the Solomon Islands arc. *Nature*, *314*(6010), 428–430. https://doi.org/10.1038/314428a0 Coyle, D. A. (1994). *The application of apatite fission track analysis to problem in tectonics. PhD thesis.* Bundoora, Victoria: La Trobe University.
- Danesi, S., Pondrelli, S., Salimbeni, S., Cavaliere, A., Serpelloni, E., Danecek, P., et al. (2015). Active deformation and seismicity in the Southern Alps (Italy): The Montello hill as a case study. *Tectonophysics*, 653, 95–108. https://doi.org/10.1016/j.tecto.2015.03.028
- Davis, D., Suppe, J., & Dahlen, F. A. (1983). Mechanics of fold-and-thrust belts and accretionary wedges. Journal of Geophysical Research, 88(B2), 1153. http://dx.doi.org/10.1029/jb088ib02p01153
- Dewey, J. F. (2005). Orogeny can be very short. Proceedings of the National Academy of Sciences, 102, 15286–15293. https://doi.org/10.1073/ pnas.0505516102
- Doglioni, C. (1992). The Venetian Alps thrust belt. In Thrust tectonics (pp. 319-324). Dordrecht: Springer.
- Ehlers, T. A. (2005). Crustal thermal processes and the interpretation of thermochronometer data. *Reviews in Mineralogy and Geochemistry*, 58, 315–350. https://doi.org/10.2138/rmg.2005.58.12
- Farley, K. A. (2000). Helium diffusion from apatite: General behavior as illustrated by Durango fluorapatite. Journal of Geophysical Research, 105, 2903–2914. https://doi.org/10.1029/1999JB900348
- Favaro, S., Schuster, R., Handy, M. R., Scharf, A., & Pestal, G. (2015). Transition from orogen-perpendicular to orogen-parallel exhumation and cooling during crustal indentation—Key constraints from 147Sm/144Nd and 87Rb/87Sr geochronology (Tauern Window, Alps). *Tectonophysics*, 665, 1–16. https://doi.org/10.1016/j.tecto.2015.08.037
- Flowers, R. M., Ketcham, R. A., Shuster, D. L., & Farley, K. A. (2009). Apatite (U–Th)/He thermochronometry using a radiation damage accumulation and annealing model. *Geochimica et Cosmochimica Acta*, 73, 2347–2365. https://doi.org/10.1016/j.gca.2009.01.015
- Fosdick, J. C., Grove, M., Hourigan, J. K., & Calderón, M. (2013). Retroarc deformation and exhumation near the end of the Andes, southern patagonia. *Earth and Planetary Science Letters*, 361, 504–517. https://doi.org/10.1016/j.epsl.2012.12.007

Frisch, W., Dunkl, I., & Kuhlemann, J. (2000). Post-collisional orogen-parallel large-scale extension in the Eastern Alps. *Tectonophysics*, 327, 239–265. https://doi.org/10.1016/S0040-1951(00)00204-3

- Frisch, W., Kuhlemann, J., Dunkl, I., & Brügel, A. (1998). Palinspastic reconstruction and topographic evolution of the Eastern Alps during late Tertiary tectonic extrusion. *Tectonophysics*, 297, 1–15. https://doi.org/10.1016/S0040-1951(98)00160-7
- Fügenschuh, B., Seward, D., & Mancktelow, N. (1997). Exhumation in a convergent orogen: The western Tauern window. Terra Nova, 9, 213–217. https://doi.org/10.1111/j.1365-3121.1997.tb00015.x
- Galadini, F., Poli, M. E., & Zanferrari, A. (2005). Seismogenic sources potentially responsible for earthquakes with $M \ge 6$ in the eastern Southern Alps (Thiene-Udine sector, NE Italy). *Geophysical Journal International*, 161(3), 739–762. https://doi.org/10.1111/j.1365-246X.2005.02571.x
- Gebrande, H., Lüschen, E., Bopp, M., Bleibinhaus, F., Lammerer, B., Oncken, O., et al. (2002). First deep seismic reflection images of the Eastern Alps reveal giant crustal wedges and transcrustal ramps. *Geophysical Research Letters*, 29, 92-1–92-4. https://doi.org/10.1029/2002GL014911
- GEOFON Data Center. (1993). GEOFON seismic network. Deutsches GeoForschungsZentrum GFZ. Other/Seismic Network. https://doi.org/10.14470/TR560404
- Grundmann, G., & Morteani, G. (1985). The young uplift and thermal history of the central Eastern Alps (Austria/Italy), evidence from apatite fission track ages. Jahrbuch der Geologischen Bundesanstait, 128, 197–216.



- Guenthner, W. R., Reiners, P. W., Ketcham, R. A., Nasdala, L., & Giester, G. (2013). Helium diffusion in natural zircon: Radiation damage, anisotropy, and the interpretation of zircon (U-Th)/He thermochronology. *American Journal of Science*, 313, 145–198. https://doi. org/10.2475/03.2013.01
- Handy, M. R., Ustaszewski, K., & Kissling, E. (2015). Reconstructing the Alps–Carpathians–Dinarides as a key to understanding switches in subduction polarity, slab gaps and surface motion. *International Journal of Earth Sciences*, 104, 1–26. https://doi.org/10.1007/s00531-014-1060-3
- Herwegh, M., Berger, A., Baumberger, R., Wehrens, P., & Kissling, E. (2017). Large-scale crustal-block-extrusion during late Alpine collision. Scientific Reports, 7(1), 1–10. https://doi.org/10.1038/s41598-017-00440-0
- Hetényi, G., Plomerová, J., Bianchi, I., Kampfová Exnerová, H., Bokelmann, G., Handy, M. R., & Babuška, V. (2018). From mountain summits to roots: Crustal structure of the Eastern Alps and Bohemian Massif along longitude 13.3°E. *Tectonophysics*, 744, 239–255. https://doi.org/10.1016/j.tecto.2018.07.001
- Houseman, G. A., McKenzie, D. P., & Molnar, P. (1981). Convective instability of a thickened boundary layer and its relevance for the thermal evolution of continental convergent belts. Journal of Geophysical Research, 86, 6115–6132. https://doi.org/10.1029/JB086iB07p06115
- Houseman, G. A., Neil, E. A., & Kohler, M. D. (2000). Lithospheric instability beneath the transverse ranges of California. Journal of Geophysical Research, 105, 16237–16250. https://doi.org/10.1029/2000JB900118
- Jozi Najafabadi, A., Haberland, C., Ryberg, T., Verwater, V., Le Breton, E., Handy, M. R., & Weber, M., & the AlpArray working group. (2020). Relocation of earthquakes in the southern and eastern alps (Austria, Italy) recorded by the dense, temporary SWATH–D network using a Markov chain Monte Carlo inversion. *Solid Earth Discuss*. (in review) https://doi.org/10.5194/se-2020-192
- Karousova, H., Plomerova, J., & Babuska, V. (2013). Upper-mantle structure beneath the southern Bohemian Massif and its surroundings imaged by high-resolution tomography. *Geophysical Journal International*, 194, 1203–1215. https://doi.org/10.1093/gji/ggt159
- Kästle, E. D., Rosenberg, C., Boschi, L., Bellahsen, N., Meier, T., & El-Sharkawy, A. (2020). Slab break-offs in the Alpine subduction zone. International Journal of Earth Sciences, 109, 1–17. https://doi.org/10.1007/s00531-020-01821-z
- Ketcham, R. A., Carter, A., Donelick, R. A., Barbarand, J., & Hurford, A. J. (2007). Improved modeling of fission-track annealing in apatite. *American Mineralogist*, 92, 799–810. https://doi.org/10.2138/am.2007.2281
- Kissling, E., Schmid, S. M., Lippitsch, R., Ansorge, J., & Fügenschuh, B. (2006). Lithosphere structure and tectonic evolution of the Alpine arc: New evidence from high-resolution teleseismic tomography. *Geological Society London Memoirs*, 32, 129–145. https://doi. org/10.1144/GSL.MEM.2006.032.01.08
- Kufner, S.-K., Schurr, B., Sippl, C., Yuan, X., Ratschbacher, L., Akbar, A. s/of M., et al. (2016). Deep India meets deep Asia: Lithospheric indentation, delamination and break-off under Pamir and Hindu Kush (Central Asia). Earth and Planetary Science Letters, 435, 171–184. https://doi.org/10.1016/j.epsl.2015.11.046
- Kummerow, J., Kind, R., Oncken, O., Giese, P., Ryberg, T., Wylegalla, K., & Scherbaum, F. (2004). A natural and controlled source seismic profile through the Eastern Alps: TRANSALP. *Earth and Planetary Science Letters*, 225, 115–129. https://doi.org/10.1016/j. epsl.2004.05.040
- Lammerer, B., Gebrande, H., Lüschen, E., & Veselá, P. (2008). A crustal-scale cross-section through the Tauern Window (eastern Alps) from geophysical and geological data. *Geological Society London Special Publications*, 298, 219–229. https://doi.org/10.1144/SP298.11
- Lippitsch, R. (2002). Lithosphere and upper mantle P-wave velocity structure beneath the Alps by high-resolution teleseismic tomography. *ETH Zurich*. https://doi.org/10.3929/ethz-a-004484684
- Lippitsch, R., Kissling, E., & Ansorge, J. (2003). Upper mantle structure beneath the Alpine orogen from high-resolution teleseismic tomography. Journal of Geophysical Research, 108(B8). https://doi.org/10.1029/2002JB002016
- Liu, T.-K., Hsieh, S., Chen, Y.-G., & Chen, W.-S. (2001). Thermo-kinematic evolution of the Taiwan oblique-collision mountain belt as revealed by zircon ¢ssion track dating. Earth and Planetary Science Letters, 12. https://doi.org/10.1016/S0012-821X(01)00232-1
- Luth, S., Willingshofer, E., Sokoutis, D., & Cloetingh, S. (2013). Does subduction polarity changes below the Alps? Inferences from analogue modeling. *Tectonophysics*, 582, 140–161. https://doi.org/10.1016/j.tecto.2012.09.028
- Lüschen, E., Borrini, D., Gebrande, H., Lammerer, B., Millahn, K., Neubauer, F., & Nicolich, R. (2006). TRANSALP—deep crustal Vibroseis and explosive seismic profiling in the Eastern Alps. *Tectonophysics, TRANSALP*, 414, 9–38. https://doi.org/10.1016/j.tecto.2005.10.014
- Lüschen, E., Lammerer, B., Gebrande, H., Millahn, K., & Nicolich, R. (2004). Orogenic structure of the Eastern Alps, Europe, from TRANSALP deep seismic reflection profiling. Tectonophysics, continental lithosphere. In Papers presented at the 10th international symposium on deep seismic profiling of the continents and their margins. (pp. 85–102). https://doi.org/10.1016/j.tecto.2004.07.024
- Manckelow, N. S., Stöckli, D. F., Grollimund, B., Müller, W., Fügenschuh, B., Viola, G., et al. (2001). The DAV and Periadriatic fault systems in the Eastern Alps south of the Tauern window. *International Journal of Earth Sciences*, 90, 593–622. https://doi.org/10.1007/ s005310000190
- McQuarrie, N., & Ehlers, T. A. (2015). Influence of thrust belt geometry and shortening rate on thermochronometer cooling ages: Insights from thermokinematic and erosion modeling of the Bhutan Himalaya. *Tectonics*, *34*, 1055–1079. https://doi.org/10.1002/2014TC003783
- McQuarrie, N., & Ehlers, T. A. (2017). Techniques for understanding fold-and-thrust belt kinematics and thermal evolution. In *Linkages and feedbacks in orogenic systems*. Geological Society of America. https://doi.org/10.1130/2017.1213(02)
 Nickel L. Ehlers, T. A. Clatchack, G. Adema, D. A. & Stillager, K. (2010). Techniques and placing and places in the systems.
- Michel, L., Ehlers, T. A., Glotzbach, C., Adams, B. A., & Stübner, K. (2018). Tectonic and glacial contributions to focused exhumation in the Olympic Mountains, Washington, USA. *Geology*, 46(6), 491–494. https://doi.org/10.1130/G39881.1
- Moratto, L., Romano, M. A., Laurenzano, G., Colombelli, S., Priolo, E., Zollo, A., et al. (2019). Source parameter analysis of microearthquakes recorded around the underground gas storage in the Montello-Collalto Area (Southeastern Alps, Italy). *Tectonophysics*, 762, 159–168. https://doi.org/10.1016/j.tecto.2019.04.030
- Most-Angelmaier, P. (2003). Report Late Alpine cooling histories of tectonic blocks along the central part of the TRANSALP-Traverse (Inntal-Gadertal): Constraints from geochronology, Ph.D. thesis. University of Tübingen.
- Müller, R. D., Zahirovic, S., Williams, S. E., Cannon, J., Seton, M., Bower, D. J., et al. (2019). A global plate model including lithospheric deformation along major rifts and orogens since the triassic. *Tectonics*, 38, 1884–1907. https://doi.org/10.1029/2018TC005462
- Müller, W., Prosser, G., Mancktelow, N. S., Villa, I. M., Kelley, S. P., Viola, G., & Oberli, F. (2001). Geochronological constraints on the evolution of the Periadriatic Fault System (Alps). *International Journal of Earth Sciences*, *90*, 623–653. https://doi.org/10.1007/s005310000187
 Nocquet, J.-M., & Calais, E. (2004). Geodetic Measurements of Crustal Deformation in the Western Mediterranean and Europe. *Pure and Applied Geophysics*, *161*, 661–681. https://doi.org/10.1007/s00024-003-2468-z
- Ortner, H., Reiter, F., & Brandner, R. (2006). Kinematics of the Inntal shear zone-sub-Tauern ramp fault system and the interpretation of the TRANSALP seismic section, Eastern Alps, Austria. *Tectonophysics, TRANSALP, 414*, 241–258. https://doi.org/10.1016/j. tecto.2005.10.017



- Petricca, P., Carminati, E., & Doglioni, C. (2019). The decollement depth of active thrust faults in Italy: Implications on potential earthquake magnitude. *Tectonics*. https://doi.org/10.1029/2019TC005641
- Petterson, M. G., Babbs, T., Neal, C. R., Mahoney, J. J., Saunders, A. D., Duncan, R. A., & Natogga, D. (1999). Geological-tectonic framework of Solomon Islands, SW Pacific: Crustal accretion and growth within an intra-oceanic setting. *Tectonophysics*, 301(1–2), 35–60. https:// doi.org/10.1016/S0040-1951(98)00214-5
- Pieri, M., & Groppi, G. (1981). Subsurface geological structure of the Po Plain. 414(1–23). Italy. Consiglio Naxionale delle Ricerche-Progetto Finalizzato Geodinamica.
- Pleuger, J., Mancktelow, N., Zwingmann, H., & Manser, M. (2012). K–Ar dating of synkinematic clay gouges from Neoalpine faults of the Central, Western and Eastern Alps. *Tectonophysics*. 550–553, 1–16. https://doi.org/10.1016/j.tecto.2012.05.001
- Pysklywec, R. N. (2001). Evolution of subducting mantle lithosphere at a continental plate boundary. *Geophysical Research Letters*, 28, 4399–4402. https://doi.org/10.1029/2001GL013567
- Pysklywec, R. N., Beaumont, C., & Fullsack, P. (2000). Modeling the behavior of the continental mantle lithosphere during plate convergence. *Geology*, 28(7), 655. http://dx.doi.org/10.1130/0091-7613(2000)28<655:mtbotc>2.0.co;2
- Ratschbacher, L., Merle, O., Davy, P., & Cobbold, P. (1991). Lateral extrusion in the eastern Alps, Part 1: Boundary conditions and experiments scaled for gravity. *Tectonics*, 10, 245–256. https://doi.org/10.1029/90TC02622
- Reiners, P. W., Ehlers, T. A., & Zeitler, P. K. (2005). Past, present, and future of thermochronology. *Reviews in Mineralogy and Geochemistry*, 58(1), 1–18. https://doi.org/10.2138/rmg.2005.58.1
- Robl, J., Heberer, B., Prasicek, G., Neubauer, F., & Hergarten, S. (2017). The topography of a continental indenter: The interplay between crustal deformation, erosion, and base level changes in the eastern Southern Alps. *Journal of Geophysical Research: Earth Surface*, 122, 310–334. https://doi.org/10.1002/2016JF003884
- Rosenberg, C. L. (2004). Shear zones and magma ascent: A model based on a review of the Tertiary magmatism in the Alps. *Tectonics*, 23. https://doi.org/10.1029/2003TC001526
- Rosenberg, C. L., & Kissling, E. (2013). Three-dimensional insight into Central-Alpine collision: Lower-plate or upper-plate indentation? Geology, 41(12), 1219–1222. https://doi.org/10.1130/G34584.1
- Rosenberg, C. L., Schneider, S., Scharf, A., Bertrand, A., Hammerschmidt, K., Rabaute, A., & Brun, J. P. (2018). Relating collisional kinematics to exhumation processes in the Eastern Alps. *Earth-Science Reviews*, 176, 311–344. https://doi.org/10.1016/j.earscirev.2017.10.013
- Scharf, A., Handy, M. R., Favaro, S., Schmid, S. M., & Bertrand, A. (2013). Modes of orogen-parallel stretching and extensional exhumation in response to microplate indentation and roll-back subduction (Tauern Window, Eastern Alps). *International Journal of Earth Sciences*, 102, 1627–1654. https://doi.org/10.1007/s00531-013-0894-4
- Schlunegger, F., & Kissling, E. (2015). Slab rollback orogeny in the Alps and evolution of the Swiss Molasse basin. *Nature Communications*, 6(1). http://dx.doi.org/10.1038/ncomms9605
- Schmid, S. M., Fügenschuh, B., Kissling, E., & Schuster, R. (2004). Tectonic map and overall architecture of the Alpine orogen. *Eclogae Geologicae Helvetiae*, 97, 93–117. https://doi.org/10.1007/s00015-004-1113-x
- Schmid, S. M., Pfiffner, O. A., Froitzheim, N., Schönborn, G., & Kissling, E. (1996). Geophysical-geological transect and tectonic evolution of the Swiss-Italian Alps. *Tectonics*, 15, 1036–1064. https://doi.org/10.1029/96TC00433
- Schmid, S. M., Scharf, A., Handy, M. R., & Rosenberg, C. L. (2013). The Tauern Window (Eastern Alps, Austria): A new tectonic map, with cross-sections and a tectonometamorphic synthesis. Swiss Journal of Geosciences, 106, 1–32. https://doi.org/10.1007/s00015-013-0123-y Schönborn, G. (1992). Kinematics of a transverse zone in the Southern Alps, Italy. In K. R. McClay, (Ed.), Thrust tectonics (pp. 299–310).
- Schonborn, G. (1992). Knematics of a transverse zone in the Southern Alps, Italy. In K. K. McClay, (Ed.), *Thrust tectorics* (pp. 299–310). Dordrecht: Springer Netherlands. https://doi.org/10.1007/978-94-011-3066-0_27
- Schönborn, G. (1999). Balancing cross sections with kinematic constraints: The Dolomites (northern Italy). *Tectonics*, 18(3), 527–545. https://doi.org/10.1029/1998TC900018
- Schwanghart, W., & Scherler, D. (2014). Topotoolbox 2–MATLAB-based software for topographic analysis and modeling in Earth surface sciences. Earth Surface Dynamics, 2(1), 1–7. http://dx.doi.org/10.5194/esurf-2-1-2014
- Selverstone, J., Axen, G. J., & Bartley, J. M. (1995). Fluid inclusion constraints on the kinematics of footwall uplift beneath the Brenner Line normal fault, eastern Alps. *Tectonics*, 14, 264–278. https://doi.org/10.1029/94TC03085
- Serpelloni, E., Vannucci, G., Anderlini, L., & Bennett, R. A. (2016). Kinematics, seismotectonics and seismic potential of the eastern sector of the European Alps from GPS and seismic deformation data. *Tectonophysics*, 688, 157–181. https://doi.org/10.1016/j.tecto.2016.09.026
- Spada, M., Bianchi, I., Kissling, E., Agostinetti, N. P., & Wiemer, S. (2013). Combining controlled-source seismology and receiver function information to derive 3-D Moho topography for Italy. *Geophysical Journal International*, 194, 1050–1068. https://doi.org/10.1093/gji/ ggt148
- Steenken, A., Siegesmund, S., Heinrichs, T., & Fügenschuh, B. (2002). Cooling and exhumation of the Rieserferner Pluton (Eastern Alps, Italy/Austria). International Journal of Earth Sciences, 91(5), 799–817. https://doi.org/10.1007/s00531-002-0260-4
- Storchak, D. A., Harris, J., Brown, L., Lieser, K., Shumba, B., Verney, R., et al. (2017). Rebuild of the Bulletin of the International Seismological Center (ISC), part 1: 1964–1979. *Geoscience Letters*, 4. https://doi.org/10.1186/s40562-017-0098-z
- Stöckhert, B., Brix, M. R., Kleinschrodt, R., Hurford, A. J., & Wirth, R. (1999). Thermochronometry and microstructures of quartz—a comparison with experimental flow laws and predictions on the temperature of the brittle–plastic transition. *Journal of Structural Geology*, 21(3), 351–369. https://doi.org/10.1016/S0191-8141(98)00114-X
- Teng, L. S., Lee, C. T., Tsai, Y. B., & Hsiao, L. Y. (2000). Slab breakoff as a mechanism for flipping of subduction polarity in Taiwan. *Geology*, 28(2), 155–158. https://doi.org/10.1130/0091-7613(2000)28<155:SBAAMF>2.0.CO;2
- Trautwein, B., Dunkl, I., & Frisch, W. (2001). Accretionary history of the Rhenodanubian Flysch zone in the Eastern Alps–evidence from apatite fission-track geochronology. *International Journal of Earth Sciences*, 90(3), 703–713. https://doi.org/10.1007/s005310000184
- U.S./Japan ASTER Science Team. 10.5067/ASTERGTM.002. see also, Abrams, M. (2000). The Advanced Spaceborne Thermal Emission and Reflection Radiometer (ASTER): Data products for the high spatial resolution imager on NASA's Terra platform. *International Journal of Remote Sensing*, 21(5), 847–859. https://doi.org/10.1080/014311600210326
- Vernon, A. J., van der Beek, P. A., Sinclair, H. D., Persano, C., Foeken, J., & Stuart, F. M. (2009). Variable late Neogene exhumation of the central European Alps: Low-temperature thermochronology from the Aar Massif, Switzerland, and the Lepontine Dome, Italy. *Tectonics*, 28. https://doi.org/10.1029/2008TC002387
- von Blanckenburg, F., & Davies, J. H. (1995). Slab breakoff: A model for syncollisional magmatism and tectonics in the Alps. *Tectonics*, 14, 120–131. https://doi.org/10.1029/94TC02051
- von Blanckenburg, F., Villa, I. M., Baur, H., Morteani, G., & Steiger, R. H. (1989). Time calibration of a PT-path from the Western Tauern Window, Eastern Alps: The problem of closure temperatures. *Contributions to Mineralogy and Petrology*, 101, 1–11. https://doi. org/10.1007/BF00387196

- Willett, S., Beaumont, C., & Fullsack, P. (1993). Mechanical model for the tectonics of doubly vergent compressional orogens. *Geology*, 21, 371–374. https://doi.org/10.1130/0091-7613(1993)021<0371:MMFTTO>2.3.CO;2
- Willett, S. D., & Brandon, M. T. (2002). On steady states in mountain belts. *Geology*, 30, 175. https://doi.org/10.1130/0091-7613(2002)030 <0175:OSSIMB>2.0.CO;2
- Willett, S. D., Herman, F., Fox, M., Stalder, N., Ehlers, T. A., Jiao, R., & Yang, R. (2020). Bias and error in modeling thermochronometric data: Resolving a potential increase in Plio-Pleistocene erosion rate. *Earth Surface Dynamics Discussions*, 1–78. https://doi.org/10.5194/ esurf-2020-59
- Willett, S. D., Slingerland, R., & Hovius, N. (2001). Uplift, shortening, and steady state topography in active mountain belts. *American Journal of Science*, 301, 455–485. https://doi.org/10.2475/ajs.301.4-5.455
- Willingshofer, E., & Cloetingh, S. (2003). Present-day lithospheric strength of the Eastern Alps and its relationship to neotectonics. *Tectonics*, 22(6). https://doi.org/10.1029/2002TC001463
- Winterberg, S., & Willett, S. D. (2019). Greater Alpine river network evolution, interpretations based on novel drainage analysis. Swiss Journal of Geosciences, 112, 3–22. https://doi.org/10.1007/s00015-018-0332-5
- Wolff, R., Hetzel, R., Dunkl, I., Anczkiewicz, A. A., & Pomella, H. (2020). Fast cooling of normal-fault footwalls: Rapid fault slip or thermal relaxation?. *Geology*, 48(4), 333–337. http://dx.doi.org/10.1130/g46940.1.
- Wölfler, A., Kurz, W., Fritz, H., & Stüwe, K., (2011). Lateral extrusion in the Eastern Alps revisited: Refining the model by thermochronological, sedimentary, and seismic data: Extrusion in the eastern alps revisited. *Tectonics*, 30. https://doi.org/10.1029/2010TC002782
- Zattin, M., Cuman, A., Fantoni, R., Martin, S., Scotti, P., & Stefani, C. (2006). From Middle Jurassic heating to Neogene cooling: The thermochronological evolution of the southern Alps. *Tectonophysics*, *TRANSALP*, 414, 191–202. https://doi.org/10.1016/j.tecto.2005.10.020
 Zattin, M., Stefani, C., & Martin, S. (2003). Detrital fission-track analysis and sedimentary petrofacies as keys of alpine exhumation: The avample of the venetion foreland (surgness poutbern alps. italy). *Journal of Sedimentary Personet*, 73, 1051–1061. https://doi.org/10.1016/j.tecto.2005.10.020
- The example of the venetian foreland (european southern alps, italy). *Journal of Sedimentary Research*, 73, 1051–1061. https://doi.org/10.1306/051403731051
- Zwingmann, H., & Mancktelow, N. (2004). Timing of Alpine fault gouges. Earth and Planetary Science Letters, 223, 415–425. https://doi.org/10.1016/j.epsl.2004.04.041

Reference From the Supporting Information

- Farley, K. A. (2002). (U-Th)/He dating: Techniques, calibrations, and applications. Reviews in Mineralogy and Geochemistry, 47(1), 819–844. https://doi.org/10.2138/rmg.2002.47.18
- Glotzbach, C., Lang, K. A., Avdievitch, N. N., & Ehlers, T. A. (2019). Increasing the accuracy of (U-Th (-Sm))/He dating with 3D grain modeling. *Chemical Geology*, 506, 113–125. https://doi.org/10.1016/j.chemgeo.2018.12.032
- Kirby, E., & Whipple, K. X. (2012). Expression of active tectonics in erosional landscapes. Journal of Structural Geology, 44, 54–75. https:// doi.org/10.1016/j.jsg.2012.07.009
- Perron, J. T., & Royden, L. (2013). An integral approach to bedrock river profile analysis. Earth Surface Processes and Landforms, 38(6), 570–576. https://doi.org/10.1002/esp.3302
- Pfingstl, S., Kurz, W., Schuster, R., & Hauzenberger, C. (2015). Geochronological constraints on the exhumation of the Austroalpine Seckau Nappe (Eastern Alps). *Austrian Journal of Earth Sciences*, 108(1). https://doi.org/10.17738/ajes.2015.0011
- Schwanghart, W., & Kuhn, N. J. (2010). TopoToolbox: A set of Matlab functions for topographic analysis. Environmental Modelling & Software, 25(6), 770–781. https://doi.org/10.1016/j.envsoft.2009.12.002